Author’s response to reviews

Title: Mobile Health in the Management of Type 1 Diabetes

Authors:

Xuemei Wang (wangxm_zsu@163.com)
Wei Shu (shuvimmu@163.com)
Jian Du (jdu-sdu@163.com)
Maolin Du (dumaolin2016@163.com)
Peiyu Wang (wpeiyu138@163.com)
Mingming Xue (happybird-nmg@163.com)
Huiqiu Zheng (zhqg1988@163.com)
Yufeng Jiang (jiangyufeng@medmail.com.cn)
Shaohua Yin (ykdynsh@163.com)
Danyan Liang (ldy77671@163.com)
Ruiqi Wang (wangruiqi@163.com)
Lina Hou (1328588962@qq.com)

Version: 4 Date: 23 Jan 2019

Author’s response to reviews:

Declare RE: BEND-D-18-00220R1, entitled "Mobile Health in the Management of Type 1 Diabetes, A systematic review"

Dear Editor:

Thank you for the detailed review of our manuscript and the helpful suggestions given by the editorial team. We have restructured manuscript and are now submitting it as synopsis, following the advice from Editorial Board. Additionally, we have worked hard to address all the comments raised and have made adjustments to our manuscript where we could. Please see below for the detailed point-by-point response. We hope you find our revised version acceptable for publication in BMC Endocrine Disorders.

Thank you again for your time and patience. Look forward to hearing from you.
Sincerely yours,

Maolin

Editor Comments:

1. We have noted that author JD is missing in the listed authors' contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship
   Response: Thanks editor for the comment. We are very sorry that we have lost the author's contribution in the revision of the manuscript. We have revised the author's contribution in accordance with the requirements of the magazine.

2. Please remove the additional files list as there are no additional files.
   Response: Thanks editor for the comment. We removed the additional files list

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
   Response: Thanks editor for the comment. We updated the manuscript.